RESEARCH ARTICLE

High seroconversion rate to Rift Valley fever virus in cattle and goats in far northern KwaZulu-Natal, South Africa, in the absence of reported outbreaks

Carien van den Bergh¹, Estelle H. Venter^{1,2}, Robert Swanepoel¹, Peter N. Thompson^{3*}

1 Department of Veterinary Tropical Diseases, Faculty of Veterinary Science, University of Pretoria, Onderstepoort, Gauteng, South Africa, 2 College of Public Health, Medical and Veterinary Sciences, James Cook University, Townsville, Queensland, Australia, 3 Epidemiology Section, Department of Production Animal Studies, Faculty of Veterinary Science, University of Pretoria, Onderstepoort, Gauteng, South Africa

* peter.thompson@up.ac.za

Abstract

Background

Rift Valley fever (RVF) is a mosquito-borne zoonotic disease characterized in South Africa by large epidemics amongst ruminant livestock at very long, irregular intervals, mainly in the central interior. However, the presence and patterns of occurrence of the virus in the eastern parts of the country are poorly known. This study aimed to detect the presence of RVF virus (RVFV) in cattle and goats in far northern KwaZulu-Natal province and to estimate the prevalence of antibodies to the virus and the incidence rate of seroconversion.

Methodology

Cross-sectional studies were performed in communally farmed cattle (n = 423) and goats (n = 104), followed by longitudinal follow-up of seronegative livestock (n = 253) 14 times over 24 months, representing 160.3 animal-years at risk. Exposure to RVFV was assessed using an IgG sandwich ELISA and a serum neutralization test (SNT) and seroconversion was assessed using SNT. Incidence density was estimated and compared using multivariable Poisson models and hazard of seroconversion was estimated over time.

Principal findings

Initial overall seroprevalence was 34.0% (95%CI: 29.5–38.8%) in cattle and 31.7% (95%CI: 22.9–41.6%) in goats, varying by locality from 18–54%. Seroconversions to RVFV based on SNT were detected throughout the year, with the incidence rate peaking during the high rainfall months of January to March, and differed considerably between years. Overall seroconversion rate in cattle was 0.59 per animal-year (95% CI: 0.46–0.75) and in goats it was 0.41 per animal-year (95% CI: 0.25–0.64), varying significantly over short distances.



G OPEN ACCESS

Citation: van den Bergh C, Venter EH, Swanepoel R, Thompson PN (2019) High seroconversion rate to Rift Valley fever virus in cattle and goats in far northern KwaZulu-Natal, South Africa, in the absence of reported outbreaks. PLoS Negl Trop Dis 13(5): e0007296. https://doi.org/10.1371/journal. pntd.0007296

Editor: Brian Bird, School of Veterinary Medicine University of California Davis, UNITED STATES

Received: October 30, 2018

Accepted: March 12, 2019

Published: May 3, 2019

Copyright: © 2019 van den Bergh et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

Funding: This project was partially supported by the Cooperative Agreement Number 5 NU2GGH001874-02-00, funded by the Centers for Disease Control and Prevention. Its contents are solely the responsibility of the authors and do not necessarily represent the official views of the Centers for Disease Control and Prevention or the Department of Health and Human Services. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Conclusions/Significance

The high seroprevalence in all age groups and evidence of year-round viral circulation provide evidence for a hyperendemic situation in the study area. This is the first study to directly estimate infection rate of RVFV in livestock in an endemic area in the absence of reported outbreaks and provides the basis for further investigation of factors affecting viral circulation and mechanisms for virus survival during interepidemic periods.

Author summary

Rift Valley fever (RVF) is a mosquito-transmitted viral disease that may cause large epidemics in domestic livestock and in humans. Although currently largely confined to Africa, it is of international concern due to its ability to spread and become established in areas where suitable mosquito vectors occur. Outbreaks occur sporadically, associated with conditions favourable for proliferation of mosquito populations, such as high rainfall and flooding, yet their location and timing remain difficult to predict. In other areas there is evidence that RVF virus is endemic and may circulate without causing outbreaks. However, the location and extent of such areas is poorly known, as is the transmission dynamics of the virus in those areas. In this paper, we report the existence of such an area of endemic RVF virus transmission on the tropical coastal plain of South Africa bordering Mozambique, where we found a high rate of exposure of domestic cattle and goats to the naturally circulating virus over a two-year period, with no outbreaks being reported. Research in such areas will help us to assess the potential for spread of the virus to other areas and also to better understand the behaviour of the virus during periods between epidemics.

Introduction

Rift Valley fever (RVF) is a zoonotic arboviral disease caused by RVF virus (RVFV), a *Phlebovirus* in the family *Phenuiviridae* [1], mainly affecting livestock and humans and transmitted by *Aedes* and *Culex* spp. mosquitoes. Non-vector transmission of RVFV is not considered important in livestock but humans are easily infected by contact with bodily fluids of infected animals or by inhaling infectious particles [2]. Human infection may present in a self-limiting febrile disease with signs including fever, severe headache, malaise, muscle pain and nausea, but in severe cases can result in encephalopathy, haemorrhagic signs, retinopathy and even death. The disease in ruminants may be characterized by necrotic hepatitis but may also be inapparent or mild, and in pregnant animals usually results in abortions and neonatal mortalities [2].

The virus was first isolated in Kenya in 1930 [3], and has been endemic in sub-Saharan Africa ever since. The virus spread beyond sub-Saharan Africa when outbreaks occurred in the Nile delta in Egypt during 1977–1978 causing mortalities in both humans and livestock [4, 5]. It was first recorded in Mauritania and Senegal in West Africa in 1987 [6] and spread beyond mainland Africa, to Saudi Arabia and Yemen in 2000 [7–9], the Comoros in 2007 [10] and Madagascar in 1990 [11]. Three major RVF epidemics have occurred in South Africa, during 1950–1951, 1974–1975 and more recently 2008–2011 [12]. During the 2010 outbreaks a total of 302 human infections were diagnosed with a case fatality rate of 8% [13] and a loss of 19,000

head of livestock was reported. Species such as buffaloes, camels and other wildlife were also affected [14]. These large epidemics have occurred mainly following unusually heavy rains on the relatively dry central plateau of South Africa. However, in the periods between these large epidemics, several smaller outbreaks or isolated cases have occurred both in the interior and in the eastern parts of the country [12].

The fate of the virus during the long interepidemic periods in South Africa is poorly understood. The virus has been isolated from newly emerged, unfed *Aedes lineatopennis (Ae. mcintoshi*) in East Africa, suggesting that it may survive extended periods in aedine mosquito eggs, which are very resilient in dry conditions [15]. Romoser et al. [16] provided laboratory evidence in support of transovarial transmission in artificially infected *Ae. mcintoshi*, and it remains the most popular theory for interepidemic survival of RVFV; however, evidence for its occurrence in nature is limited. Other possibilities are the occurrence of low-level interepidemic circulation of virus between vectors and unknown hosts in endemic areas, or disappearance of virus from an area with later re-introduction from an endemic area via movement of infected hosts or vectors [2].

One such endemic area may be the low-lying, tropical eastern part of southern Africa, including Mozambique and small parts of north-eastern South Africa. Very few outbreaks have been reported in Mozambique, yet there is serological evidence of widespread exposure to RVFV in livestock [17] and humans [18]. RVFV was responsible for a small outbreak in goats in the Maputo Province, close to the South African border, in 2014 [19]. Recently, the overall seroprevalence of RVFV antibodies in seven provinces of Mozambique was reported to be 37% in cattle, 30% in African buffalo (*Syncerus caffer*), and 29% in domestic ruminants in the Maputo Province [20]. In South Africa, RVFV was isolated from *Ae. circumluteolus* in the tropical coastal region of far northern KwaZulu-Natal (KZN) in 1955 [21] and at the same time neutralizing antibodies to RVFV were found in 12% of domestic livestock [22] and 10% of humans [23]. However, subsequent efforts to detect the virus in mosquitoes in that region were unsuccessful [24]. There have also not been any reports of clinical cases of RVF in that region, although further south along the KZN coast there was some evidence of RVFV circulation and sporadic clinical cases in 1972–1973 [25] and in a dairy herd in 1981 [26].

Various reports have inferred the occurrence of interepidemic circulation of RVFV on the basis of seropositivity in the absence of reported outbreaks or vaccination, in livestock [17, 27, 28] wildlife [29, 30] and humans [31, 32]. Two recent studies have reported longitudinal follow-up of sentinel sheep and goats in Kenya, reporting low rates of seroconversion in areas that had previously experienced outbreaks [33, 34]. However, no published studies have attempted to quantify the rate of seroconversion in livestock by follow-up of seronegative animals in a region where outbreaks have never been reported.

The objectives of this study were to determine whether there was evidence of recent circulation of RVFV in the tropical region of far northern KwaZulu-Natal, South Africa, to estimate the seroprevalence in cattle and goats, and to estimate the incidence rate and patterns of seroconversion in domestic ruminants over a two-year period.

Materials and methods

Ethics statement

The study protocol was approved by the Animal Ethics Committee of the University of Pretoria (V013-16) and the Department of Agriculture, Forestry and Fisheries, Republic of South Africa, and adhered to the specifications of the South African National Standard (SANS 10386–2008): "The Care and Use of Animals for Scientific Purposes". Support for the project was obtained from the Director of Veterinary Services of KwaZulu-Natal and the State Veterinarian: Jozini. Verbal consent was obtained from animal owners after the aims of the study were explained to them by the State-employed animal health technicians. Farmers were rewarded for their participation by providing them with small amounts of animal health products such as wound aerosol, topical treatments and endoparasiticides.

Study area

The study was conducted on the Maputaland coastal plain in far northern KwaZulu-Natal just south of the Mozambique border, an area with a tropical climate [35] characterised by warm, dry winters and hot, wet summers (Fig 1). In the Pongolo River floodplain, the core of the study area, the summer temperatures range from 23–40°C, winter temperatures range from 16–26°C and mean annual rainfall is 600–800 mm [36]. The northwards-flowing Pongolo River forms a floodplain at an elevation of 30–50 m above sea level, approximately 10–15 km to the east of the foothills of the Lebombo Mountains, which rise to about 650 m above sea level. The study area contains a mosaic of bushland, thicket, wooded grassland, riverine forest and floodplain vegetation [37]. Depending on rainfall, seasonal floodplain, filling numerous pans, some of which retain water during the dry season [38]. The construction of the Pongola-poort Dam in 1973 largely disrupted this cycle, with flooding only occurring with heavy rainfalls or with periodic opening of sluice gates. In very dry years, permanent water may only be present in the Pongolo and Usuthu Rivers and in very few of the larger pans. This study started during a drought period when most of the pans were dry (Fig 2).



Fig 1. A map of the study area showing major rivers and temporary wetlands, with the locations of the nine diptanks where cattle were sampled and the area where goats were sampled for the cross-sectional study. The map was constructed for this manuscript in Esri ArcGIS 10.2 using country boundaries from Esri ArcGIS Online, diptank coordinates collected during the study, and river [39], wetland [40] and protected area boundary [41] data available under a Creative Commons Attribution (CC BY 4.0) license.

https://doi.org/10.1371/journal.pntd.0007296.g001



Fig 2. Actual monthly rainfall recorded at Ndumo Game Reserve (NGR) in the study area, with 30-year average. Totals above the bars indicate rainfall for each year (July–June). The blue arrow indicates the study period (June 2016–June 2018).

https://doi.org/10.1371/journal.pntd.0007296.g002

Livestock in the study area (mainly cattle and goats) belong to numerous small-scale farmers and are communally grazed. Animals range widely during the day, during the dry season foraging mainly in the floodplains, and may be confined in pens at night. Much of the area lies within a foot-and-mouth disease (FMD) protection zone and cattle are brought weekly (summer) or every two weeks (winter) to diptanks where they are inspected by animal health technicians for signs of FMD and plunge-dipped for ectoparasite control using amitraz. For this inspection purpose, most cattle are individually identified using ear tags. The only vaccinations applied are those provided by the State, namely brucellosis, anthrax, blackquarter and sometimes lumpy skin disease vaccines.

Study design and sampling

The study took the form of an initial cross-sectional survey of individually identified cattle at nine diptanks (Fig 1) during June 2016, followed by a longitudinal follow-up of seronegative cattle to detect and quantify seroconversion. The nine diptanks were selected on the basis of

their location adjacent to the floodplain of the Pongolo River and its tributaries (seven diptanks) and the eastern coastal plain (two diptanks), as well as the willingness of farmers and animal health technicians to participate.

Of the seven diptanks in the western part of the study area, all were situated within 1.5 km of the Pongolo River, except for Mpala which was adjacent to a small tributary which held water for almost the entire study period and Ndumo which was about 4 km from the Pongolo River and adjacent to the Ingwavuma River which remained dry throughout the study period. The two diptanks in the eastern part of the study area were situated in a very sandy area with a mosaic of grassland and bush and with numerous small temporary pans during the wet season.

Sample size (*n*) was calculated in order to detect a seroprevalence (P_{exp}) of 25% with an allowable error (*d*) of 10% and 95% confidence, using the formula $n = 1.96^2 P_{exp}(1 - P_{exp}) / d^2$ [42]. The required sample size of 73 was then multiplied by a design effect (*D*) of 5.9, calculated by assuming an intracluster correlation coefficient (ρ) of 0.1 and a cluster size (*m*) of 50 animals per diptank, using the formula $D = 1 + \rho(m-1)$ [43]; the overall required sample size was therefore 432.

Animals were selected using systematic random sampling, selecting 2–5 animals per farmer at each diptank as they passed through a crush either before or after dipping. Only cattle >6 months old were sampled. Following serological testing of samples from the cross-sectional study, as many seronegative animals as possible, as well as a small number of seropositive animals, were then re-sampled during 13 sampling episodes at 1-2-month intervals between November 2016 and June 2018. To compensate for loss to follow-up, some additional cattle from the same owners were recruited at various times during the study.

A smaller study of the same type was conducted in goats in an area close to one of the diptanks (Fig 1); 104 goats belonging to seven farmers were sampled during February to April 2017 and seronegative animals were re-sampled at 1–2 monthly intervals until June 2018.

Blood samples were collected from the jugular vein of goats and coccygeal vein of cattle. Samples were refrigerated and transported to a BSL2+ laboratory at the Faculty of Veterinary Science, University of Pretoria, where they were centrifuged. The sera were inactivated at 56°C for 1 hour and stored at -20°C until used.

Serology

All cattle serum samples that were collected for the cross-sectional study were tested for RVFV IgG antibodies using an IgG sandwich enzyme-linked immunosorbent assay (ELISA) with reported sensitivity of 96.3% and specificity of 99.7% in cattle [44]. In addition, approximately every second sample (due to economic constraints) was tested using the serum neutralization test (SNT). Where possible, samples that tested positive for RVFV IgG were also tested for IgM antibodies using an IgM capture ELISA with reported sensitivity of 100% in sheep and specificity of 99.2% in cattle [44]. In order to detect seroconversion, all samples collected during the longitudinal study, as well as all goat serum samples, were tested using SNT.

Heat-inactivated serum samples were tested by the IgG sandwich ELISA using a previously described and validated method [44]. F96 Maxisorp Immunoplates (AEC-Amersham) were coated with anti-RVF hyperimmune mouse ascitic fluid. Optical densities (OD) were measured at 450 nm. The net OD values were first recorded for each serum as the value determined with the RVFV antigen minus the value determined with the control antigen and subsequently converted into percentage of the OD value of a high positive control serum. A cut-off value optimized for this assay (PP value \geq 30) was used.

The IgM capture ELISA was performed following a method previously described [44]. Briefly: plates were coated overnight at 4° C with 100 µl rabbit anti-sheep IgM. OD was

determined at 450 nm. The PP was determined as above and an optimized cut-off value of PP \geq 30 was used.

The SNT was performed in 96-well plate (AEC-Amersham) format according to the standard protocol of the Virology Section, Department of Veterinary Tropical Diseases, Faculty of Veterinary Science, University of Pretoria, which follows the method prescribed by the World Organisation for Animal Health [45]. Briefly: Sera were diluted 1:5 in PBS+ (phosphate buffered saline with added MgCl and CaCl) and two-fold dilutions of the serum were made. The TCID₅₀ was determined using the Karber method [46]. A volume of 100 TCID₅₀ virus (Smithburn vaccine strain) was added to each dilution and incubated at 37°C for 60 min. A total of 80 µl of African green monkey kidney cells (Vero) (480,000 cells/mL) in MEM containing 5% foetal calf serum (Biowest, Celtic) was added to each well. The microplates were incubated at 37°C in an atmosphere containing 5% CO₂ and observed daily for cytopathic effect; the titre was taken as the dilution at which 50% of the cells were affected. Results were only accepted if all controls gave the expected results (virus control, positive serum and negative serum). A serum dilution of \geq 1:10 was used to define seropositivity.

Statistical analysis

For samples tested during the cross-sectional study using both the IgG ELISA and the SNT, sensitivity and specificity of the IgG ELISA were estimated, with exact binomial 95% confidence intervals, using the SNT as the gold standard. For animals on which both tests were used, the SNT result was used to define seropositivity, otherwise the IgG ELISA result was used. The RVFV seroprevalence in cattle in June 2016 was calculated overall and by diptank, age group (≤ 2 y, 2.5–3.5 y, 4–5.5 y, ≥ 6 y) and sex. For goats the seroprevalence in February 2017 was calculated overall and by age group and sex. Exact binomial 95% confidence intervals were calculated and seroprevalence was compared between diptanks (cattle only), age groups and sexes using multiple logistic regression to control for confounding.

The occurrence of seroconversion was defined as a seronegative result (either during the cross-sectional study as defined above, or during subsequent testing by SNT) followed by a sample from the same animal that tested positive by SNT at a serum dilution of 1:10 or greater. The incidence rate of seroconversion over the 24-month study period (16 months in goats) was then calculated by including all animals that were sampled more than once (irrespective of date of first sampling) and that were seronegative on the first sampling. For the animals that did not seroconvert, animal-time at risk was calculated as the interval between the first and last tests. For animals that did seroconvert, the date of seroconversion was unknown, but was assumed to be the midpoint between the last negative and first positive test. The seroconversion rate was calculated as the number of seroconversions divided by the total animal-years at risk observed, with Poisson exact 95% confidence limits. Estimates were calculated overall for each species, separately for each year of the study (June 2016 to June 2017 and July 2017 to June 2018) and for cattle also separately for each diptank. Associations of age, sex and place with incidence rate were then assessed by including these variables in a multiple Poisson regression model, using animal-time at risk as the exposure variable.

For each species, the seroconversion rate with 95% confidence interval was plotted over time as the derivative of the kernel-smoothed Nelson-Aalen cumulative hazard estimator, using a 60-day kernel width. For this purpose, instead of the midpoint-imputed seroconversion date used above, a randomly selected date between the last negative and first positive test was used for seroconverting animals in order to avoid systematic bias in incidence rate estimation over time [47]. All statistical analyses were done using Stata 15 (StataCorp, College Station, TX, U.S.A.) and statistical significance was assessed at *P*<0.05. Risk surfaces for

seroprevalence and seroconversion rate in the study area were created by interpolation using ordinary kriging in ArcGIS 10.2 (Esri, Redlands, CA, U.S.A.).

Results

A total of 423 cattle was sampled from nine diptanks in June 2016, of which 131 (31.0%) were positive by IgG ELISA. Of 241 cattle tested using SNT, 87 (36.1%) were positive; of these, 75 were IgG ELISA-positive, giving an estimated sensitivity for the IgG ELISA of 86.2% (95% CI: 77.1–92.7%). Of 154 SNT-negative animals, 152 were IgG ELISA-negative, giving an estimated specificity of 98.7% (95% CI: 95.4–99.8%).

Seroprevalence

The seroprevalence in cattle sampled from nine diptanks in June 2016, based on IgG ELISA and SNT, was 34.0% (144/423; 95% CI 29.5–38.8%). Seroprevalence within diptanks varied from 18% to 54% and it was high in all age groups (Table 1). The multiple logistic regression model (S1 Table) showed significant differences between diptanks, with the odds of seropositivity at Mbangwini (Odds ratio (OR) = 4.5; 95% CI: 1.8–11; P = 0.001), Phelandaba (OR = 4.6; 95% CI: 1.8–12; P = 0.001) and Shemula (OR = 5.8; 95% CI: 2.3–15; P < 0.001) being higher than at Madlakude. Odds of seropositivity was the lowest in the 2–4 y age group, being significantly higher in the 4–6 y age group (OR = 2.4; 95% CI: 1.3–4.4; P = 0.006), tending to be higher in animals >6 y (P = 0.095), and tending to be higher in females than in males (P = 0.064). Of the 144 seropositive animals, 114 were tested using the IgM ELISA and 5 (4.4%) were positive; this included 2/6 (33%) of samples that were SNT-positive but IgG ELISA-negative. Therefore the estimated prevalence of IgM-positive animals in the general population was 1.3% (5/393; 95% CI: 0.4–2.9%).

Variable	n	Seroprevalence (%)*	95% CI
Diptank			
Madlakude	50	18	7–29
Masondo	21	19	2-36
Mpala	52	25	13-37
Namaneni	50	26	14-38
Ndumo	50	30	17-43
Hlanjwana	50	34	21-47
Phelandaba	50	44	30-58
Mbangwini	50	48	34-62
Shemula	50	54	40-68
Age (years)			
<2	56	34	22-46
2 -<4	96	23	15-31
4-6	157	39	31-47
>6	114	34	25-43
Sex			
female	331	36	31-41
male	92	27	18-36
Total	423	34.0	29.5-38.0

Table 1. Rift Valley fever seroprevalence in cattle at diptanks in far northern KwaZulu-Natal, June 2016.

* Based on IgG ELISA, confirmed by serum neutralization test in 241/423 samples.

https://doi.org/10.1371/journal.pntd.0007296.t001

The seroprevalence in goats sampled during February—April 2017, determined using SNT, was 31.7% (33/104; 95% CI: 22.9–41.6%) (Table 2). Although seroprevalence appeared to increase with age, this was not statistically significant (P = 0.549) in the multiple logistic regression model (S2 Table), nor was there a significant difference between sexes (P = 0.139). The overall seroprevalence in cattle and goats varied substantially across the study area (Fig 3), with foci of high seroprevalence along the Pongolo River, but was not clearly linked to the presence of wetlands.

Seroconversion rate

Of the 279 cattle that were seronegative in June 2016 by IgG ELISA, or by ELISA and SNT if both tests were used, 103 were re-sampled between one and eight times during the study period. An additional 91 SNT-seronegative cattle were recruited at various times during the study period and re-sampled at least once, therefore 194 initially seronegative animals were sampled at least twice over periods ranging between 28 and 721 days. Assuming that seroconversions occurred, on average, midway between the last negative and first positive test, a total of 111.8 animal-years at risk were observed. Due to various factors, including logistical reasons and varying farmer compliance, the majority of animals (167/194) were followed up at four of the diptanks (Namaneni, Mpala, Shemula and Ndumo), with very few followed up at Hlanjwana (13), Madlakude (8) and Masondo (6) and none at Mbangwini and Phelandaba.

Seroconversion, determined by SNT, occurred at some stage during the study period in 66 (34%) of initially seronegative cattle. Seroconversions were observed during every interval between samplings for the duration of the study, and at every diptank followed except Masondo, where only six animals were followed for five months each. The overall incidence rate of seroconversion in cattle between June 2016 and June 2018 was 0.59 seroconversions per animal-year (95% CI: 0.46–0.75) (Table 3). Seroconversion rate varied between diptanks, with the lowest rate recorded at Namaneni (0.35 per animal-year) and the highest rates observed at Hlanjwana (1.50 per animal-year) and Ndumo (1.01 per animal-year) (Fig 4). There was no clear relationship at diptank level between the initial seroprevalence in 2016 and the subsequent rate of seroconversion, and the correlation between the two outcomes was not significant (Spearman's $\rho = 0.29$; P = 0.535).

There was a marked variation in the estimated rate of seroconversion in cattle over time, with the rate being lower during the second year of the study (Table 3). Seroconversion rate increased from October 2016 to peak at just over 1.5 per animal-year between February and April 2017, dropping to almost zero during August-September 2017 and then showing a lower peak of just over 0.5 per animal-year during December-January 2018 (Fig 5).

Variable	n	Seroprevalence (%)*	95% CI
Age (years)			
0.5-1.5	25	24	9-45
>1.5-3.5	36	28	14-45
>3.5	43	40	25-56
Sex			
female	70	36	25-48
male	28	18	6-37
Total	104	31.7	22.9-41.6

Table 2.	Rift Vallev f	ever seroprevale	ence in goats in f	ar northern KwaZul	u-Natal, February-April 2017.

* Based on serum neutralization test.

https://doi.org/10.1371/journal.pntd.0007296.t002



Fig 3. Rift Valley fever virus seroprevalence in cattle and goats in far northern KwaZulu-Natal determined using IgG ELISA and serum neutralization test: risk surfaces created by interpolation using ordinary kriging. The map was constructed for this manuscript in Esri ArcGIS 10.2 using country boundaries from Esri ArcGIS Online, diptank coordinates collected during the study, and river [39], wetland [40] and protected area boundary [41] data available under a Creative Commons Attribution (CC BY 4.0) license.

https://doi.org/10.1371/journal.pntd.0007296.g003

CATTLE	Year 1 (June	2016 – June 201	7)	Year 2 (July	2017 –June 2018	8)		Total	
	Animal-years at risk†	Incidence rate	95% CI	Animal-years at risk†	Incidence rate	95% CI	Animal-years at risk†	Incidence rate	95% CI
Diptank									
Mpala	16.87	0.83	0.45-1.39	11.09	0.18	0.02-0.65	27.96	0.57	0.33-0.93
Ndumo	11.17	1.34	0.75-2.21	6.64	0.45	0.09-1.32	17.81	1.01	0.60-1.60
Namaneni	16.44	0.36	0.13-0.79	15.00	0.33	0.11-0.78	31.44	0.35	0.17-0.63
Shemula	13.74	0.73	0.35-1.34	10.64	0.19	0.02-0.68	24.37	0.49	0.25-0.86
Hlanjwana	3.25	1.84	0.68-4.01	0.75	0.00	0.00-4.90	4.01	1.50	0.55-3.26
Madlakude	3.74	0.80	0.17-2.35	0	-	-	3.74	0.80	0.17-2.35
Masondo	2.41	0.00	0.00-1.53	0	-	-	2.41	0.00	0.00-1.53
Total	67.62	0.80	0.60-1.04	44.13	0.27	0.14-0.47	111.75	0.59	0.46-0.75
GOATS	Year 1 (Feb	ruary–June 2017	ry-June 2017) Year 2 (July 2017 – June 2018)		8)	Total			
	Animal-years at risk†	Incidence rate	95% CI	Animal-years at risk†	Incidence rate	95% CI	Animal-years at risk†	Incidence rate	95% CI
Total	14.20	0.92	0.49-1.57	34.32	0.20	0.08-0.42	48.52	0.41	0.25-0.64

Table 3. Incidence rate of seroconversion^{*} to Rift Valley fever virus in cattle and goats in far northern KwaZulu-Natal between June 2016 and June 2018, expressed as numbers of seroconversions per animal-year.

* Seroconversion defined as a previously seronegative animal becoming seropositive based on the serum neutralization test at a serum dilution of 1:10 or greater. † Seroconversion assumed to have occurred midway between the last negative and the first positive test.

https://doi.org/10.1371/journal.pntd.0007296.t003



Fig 4. Rift Valley fever virus seroconversion rate in cattle and goats in far northern KwaZulu-Natal between June 2016 and June 2018, determined by serum neutralization test: Risk surfaces created by interpolation using ordinary kriging. The map was constructed for this manuscript in Esri ArcGIS 10.2 using country boundaries from Esri ArcGIS Online, diptank coordinates collected during the study, and river [39], wetland [40] and protected area boundary [41] data available under a Creative Commons Attribution (CC BY 4.0) license.

https://doi.org/10.1371/journal.pntd.0007296.g004





https://doi.org/10.1371/journal.pntd.0007296.g005



Fig 6. Seroconversion rate of goats to Rift Valley fever virus in far northern KwaZulu-Natal between February and December 2017, determined by serum neutralization test and expressed as numbers of seroconversions per animal-year, plotted using the derivative of the kernel-smoothed Nelson-Aalen cumulative hazard estimator.

https://doi.org/10.1371/journal.pntd.0007296.g006

Of the goats that were initially seronegative, 95 were followed up for periods of time ranging between 28 and 482 days, of which 20 (21%) seroconverted. A total of 48.5 animal-years at risk were observed. The overall incidence rate of seroconversion in goats between February 2017 and June 2018 was 0.41 seroconversions per animal-year (95% CI: 0.25–0.64) (Table 3) and was also markedly lower during the second summer of the study (Fig 6). The seroconversion rate is not shown after December 2017 since no further seroconversions were observed in the study animals.

For cattle and goats combined, the overall rate of seroconversion was 0.54 per animal-year (95% CI: 0.43–0.66), and was 0.82 per animal-year (95% CI: 0.63–1.04) during the first year and 0.24 per animal-year (95% CI: 0.15–0.38) during the second year. The associations of species, sex, study year and place with seroconversion rate in the multiple Poisson regression model are shown in Table 4. The rate did not differ significantly between species and sexes. However, the rate during the second study year was only a third of that during the first year (P<0.001), and significant differences were observed between locations, with Ndumo and Hlanjwana showing higher rates than Namaneni, where the lowest rate was recorded.

Discussion

This study showed evidence of a high level of exposure to RVFV in the study area, demonstrating active circulation of RVFV in livestock not associated with reported outbreaks, and to our knowledge is the first published study to quantify directly the seroconversion rate in livestock using the gold-standard SNT. Previous studies to detect interepidemic circulation have inferred low-level viral transmission in epidemic-prone areas by ELISA-based detection of IgG or IgM.

The study was done in an area of South Africa where outbreaks of RVF have never been reported, although occasional small outbreaks and suspected cases had occurred in the more temperate inland and coastal areas to the south of the study area between 1972 and 1986 [12, 25, 26]. Elsewhere in Africa, endemic circulation of RVFV is known or suspected to occur in

Variable and level	Incidence rate ratio	95% CI	P-value
Species			
Cattle	1*	-	-
Goats	1.62	0.77-3.41	0.206
Sex			
Female	1*	-	
Male	0.64	0.36-1.14	0.131
Year			
2016-2017	1*	-	-
2017-2018	0.31	0.18-0.53	< 0.001
Place			
Namanini	1*	-	-
Mpala	1.48	0.69-3.19	0.318
Ndumo	2.39	1.11-5.13	0.025
Hlanjwana	3.01	1.10-8.21	0.031
Madlakude	1.46	0.40-5.26	0.566
Shemula	1.40	0.62-3.19	0.417
Masondo	0.00	0.00 - ∞	0.977

Table 4. Factors associated with incidence rate of seroconversion to Rift Valley fever virus in cattle and goats in far northern KwaZulu-Natal between June 2016 and June 2018.

* Reference level

https://doi.org/10.1371/journal.pntd.0007296.t004

many areas. Limited numbers of RVFV infections were confirmed to occur in sheep and cattle every year during a seven year interepidemic period in Zimbabwe [48]. A seroprevalence of 33% in cattle and 26% in goats was reported during an interepidemic period in outbreakprone Garissa County, Kenya during 2013, six years after the last major outbreak [49]. Along the Nile River in Sudan, where outbreaks also occur, a prevalence of 33% in cattle and 34% in goats was found [50]. In Mauritania, also in an area that had experienced outbreaks, a seroprevalence of 15% was detected in cattle and 4% in small ruminants, and IgM and viral RNA was detected in one bovine [51]. A series of surveys in Mayotte found seroprevalences of up to 37% in cattle in the absence of reported outbreaks [52]. A seroprevalence of 17% was reported in Rwanda, also in the absence of reported outbreaks [53]. In Maputo Province, Mozambique, neighbouring our study area to the north, a survey in 2010-2011, during the large outbreak in the central interior of South Africa, found a seroprevalence in cattle of 37%, similar to that in our study [54]. This was followed in 2014 by the detection of a small outbreak of abortion in goats due to RVFV, less than 100 km north of our study area and in the same ecological zone [19]. The 1.3% IgM seroprevalence in this study was comparable to that reported in other endemic areas, e.g. 0.5% in Tanzania [28] and 0-4% in Mayotte [52], indicating the presence of low numbers of recent infections. The seroprevalence in the current study was therefore similar to or higher than those reported elsewhere in Africa where interepidemic circulation is known or suspected to occur and is consistent with ongoing or endemic circulation of RVFV with low numbers of clinical cases likely occurring but overlooked, misdiagnosed or unreported.

Seroprevalence was high in all age groups throughout the study area. Numerous studies have shown that seroprevalence is usually higher in older animals [17, 28, 55, 56], although in northern Somalia it was higher in younger animals [57]. The high seroprevalence in young animals can be a result of a recent outbreak; however, it is unlikely that a large, extensive outbreak would have gone unnoticed. In addition, rainfall in the study area during 2014–2016

had been well below average and the study area, as well as much of South Africa had experienced the worst drought in 23 years [58]. Release of water from the Pongolapoort Dam had not occurred and most of the floodplain pans were empty. Nevertheless, permanent water was still present in the major rivers and some pans, providing breeding habitat for the mosquito vectors. It is therefore likely that the high seroprevalence in all age groups is indicative of a hyperendemic situation [59] with continuous or sporadic viral activity despite the dry conditions, and including recent viral circulation.

Rate or force of infection is a key parameter in modelling of infectious diseases in an attempt to understand the factors determining disease transmission. Our study showed a high seroconversion rate to RVFV in livestock, even during dry conditions, with substantial spatial and temporal variation. Low levels of seroconversion in sentinel sheep and goats have recently been reported in Kenya in areas in which outbreaks are known to occur [33, 34]; however, positive tests were not confirmed using virus neutralization and incidence rates were not reported. In the Kruger National Park, South Africa, seroconversion was demonstrated in 5/227 African buffalo (Syncerus caffer), indicating a low annual incidence rate of 1-3% [60]. This was also in the eastern part of the country, adjacent to Mozambique, but outside the tropical zone and in a wildlife area with no livestock and a very low human population. The epidemiology of RVFV in our study area more likely approximates that in other tropical areas of Africa and therefore our results provide useful insight into the dynamics of RVFV transmission in endemic areas, both during interepidemic periods in areas where outbreaks are known to occur, and in areas where outbreaks are not reported. Specifically, the range of seroprevalence and seroconversion rates observed in our study will inform the design of further epidemiological studies in known or suspected endemic areas, as well as the modelling of RVFV transmission in such ecosystems. Knowledge of likely force of infection in endemic areas, and its seasonal variation, will also inform studies to assess the risk of spread of RVFV from such areas via animal movements.

As expected, the highest infection rate was seen during mid to late summer, when water levels are highest and mosquitoes are most abundant. However, despite the fact that rainfall was much higher during the second year of the study, the seroconversion rate was markedly lower, approximately one third of that observed during the first year. It is possible that the drought over the 1-2-year period prior to the study resulted in lower than normal infection rates with a resultant increased number of susceptible animals. This would have increased the number of viraemic animals available to infect mosquitoes, resulting in a higher than usual infection rate during the first year, followed by fewer viraemic animals during the second year. However, the suitable breeding habitat for vectors depends not only on local rainfall, but also on water levels in rivers that are determined by rainfall in their catchment areas. The important mosquito vector species in the study area, and their host preferences, are currently poorly known, and many other factors likely determine their population dynamics, most of which are poorly understood. This study showed very little difference in seroprevalence and seroconversion rate between cattle and goats, suggesting that RVFV is transmitted by one or more mosquito species without strict host preferences.

The differences in seroprevalence and in seroconversion rates between locations could not be readily explained based on geographic location or environmental factors. Two diptanks with high seroprevalence and seroconversion rates (Mbangwini and Hlanjwana, respectively) were close together and near the lower (northern) Pongolo River; however, they could not be distinguished from the other locations close to the river based on environmental or climatic variables and separate rainfall figures were not available for each location; animal management practices were also similar for all locations. Therefore, the risk and rate of RVFV infection are likely determined by a complex interplay of host, vector and environmental factors which our results show may vary substantially within a region, even over small distances. This is consistent with the marked spatial variation in seroprevalence observed in other studies [34, 52, 54, 56]. The lack of an association between seroprevalence (reflecting past seroconversion rate) and subsequent seroconversion rate in our study, as well as the differences in seroconversion rates between the first and second years of the study, also indicates that these factors vary substantially over time. More detailed, fine-scale eco-epidemiological studies to describe the dynamics of these virus-vector-host-environment interactions are required in order to better understand the behaviour of and risk posed by RVFV in such endemic ecosystems.

The seroprevalence in livestock found in this study was much higher than the 12% observed in 1955, the only previous study done in this region [22]. Since then the human and livestock populations in the area have increased tremendously, resulting in a substantial increase in RVFV host density, and it is possible that this has changed the dynamics of viral circulation. Nevertheless, the absence of sheep, the most susceptible livestock species, and the extensive farming system with year-round breeding may be responsible for the fact that large outbreaks do not occur in the area. However, sporadic cases of abortion and neonatal mortality are unlikely to be reported and investigated and may be ascribed to other causes; in addition, disease surveillance in the area is sub-optimal and access to veterinary services is difficult. Therefore, the contribution of RVF to livestock morbidity in the area is unknown and requires further investigation.

Likewise, the potential impact of RVF on human health in the area is unknown. It is generally considered that human infection by RVFV in southern Africa occurs mainly via contact with infected animal tissues from clinical cases and that infection from mosquito bites is less common, as most human cases have a history of exposure to infected animal tissues and the known mosquito vectors tend to be zoophilic [2]. Therefore, the lack of reported cases in livestock in our study area suggest that human exposure to the virus may be low. However, the last published serological surveys done in this region were in the 1950's, in which 16% of human sera (19/118) collected from locations within in our present study area were protective against RVFV [23]. To our knowledge there has never been a report of clinical RVF in humans in the area; however, the fact that malaria is common there may result in other causes of febrile illness being overlooked. In support of this hypothesis, a recent study in Maputo, 100 km north of our study area, found a 10% RVFV IgG seroprevalence in febrile patients at a primary health care facility, with seroconversion to RVFV in 5%, concluding that undiagnosed RVFV infections occur in that region and that most RVF cases are misdiagnosed as malaria [61]. Therefore, the current status and potential impact of RVFV infection in humans in our study area, and in similar areas along the coastal plain of southern Africa, urgently require further investigation.

Conclusion

The results of this study show the presence of RVFV antibodies and active seroconversion in domestic ruminants in far northern KwaZulu-Natal, indicating that RVFV is circulating in the area at a rate that varies by location, season and year. Further investigation should be done into the important mosquito vectors in the area, the factors affecting viral circulation and survival, and the impact of the presence of RVFV on animal and human health.

Supporting information

S1 Data. Dataset for cross-sectional and longitudinal studies. (XLSX)

S1 Table. Final multiple logistic regression model of factors associated with seropositivity to Rift Valley fever virus in cattle at diptanks in far northern KwaZulu-Natal, June 2016. (DOCX)

S2 Table. Multiple logistic regression model of factors associated with seropositivity to Rift Valley fever virus in goats in far northern KwaZulu-Natal, February-April 2017. (DOCX)

Acknowledgments

The authors wish to acknowledge the kindness and patience of the farmers and herders in the study area, as well as the generous assistance of the State Veterinarian and the Animal Health Technicians of the KwaZulu-Natal Department of Agriculture and Rural Development, Jozini District, and Ezemvelo KZN Wildlife. We thank Ezemvelo KZN Wildlife for supplying the weather data for Ndumo Game Reserve. We are extremely grateful to Yusuf, Dannet, Tinashe, Bruce, Christa and Ginette for help with sample collection and to Karen Ebersohn for excellent technical assistance in the laboratory.

Author Contributions

Conceptualization: Peter N. Thompson.

Data curation: Carien van den Bergh, Peter N. Thompson.

Formal analysis: Carien van den Bergh, Peter N. Thompson.

Funding acquisition: Estelle H. Venter, Robert Swanepoel, Peter N. Thompson.

Investigation: Carien van den Bergh, Peter N. Thompson.

Methodology: Carien van den Bergh, Estelle H. Venter, Robert Swanepoel, Peter N. Thompson.

Project administration: Carien van den Bergh, Peter N. Thompson.

Resources: Estelle H. Venter, Robert Swanepoel, Peter N. Thompson.

Supervision: Estelle H. Venter, Robert Swanepoel, Peter N. Thompson.

Writing - original draft: Carien van den Bergh, Peter N. Thompson.

Writing – review & editing: Carien van den Bergh, Estelle H. Venter, Robert Swanepoel, Peter N. Thompson.

References

- Adams MJ, Lefkowitz EJ, King AM, Harrach B, Harrison RL, Knowles NJ, et al. Changes to taxonomy and the International Code of Virus Classification and Nomenclature ratified by the International Committee on Taxonomy of Viruses (2017). Arch Virol. 2017 Aug 1; 162(8):2505–38. https://doi.org/10. 1007/s00705-017-3358-5 PMID: 28434098. https://doi.org/10.1007/s00705-017-3358-5.
- Swanepoel R, Coetzer JAW. Rift Valley fever. In: Coetzer JAW, Tustin RC, editors. Infectious Diseases of Livestock. Cape Town: Oxford University Press; 2004. p. 1037–70.
- Daubney R, Hudson JR, Garnham PC. Enzootic hepatitis or Rift Valley fever. An undescribed virus disease of sheep cattle and man from East Africa. J Pathol Bacteriol. 1931; 34(4):545–79. https://doi.org/ 10.1002/path.1700340418.
- Laughlin LW, Meegan JM, Strausbaugh LJ, Morens DM, Watten RH. Epidemic Rift Valley Fever in Egypt: observation of the spectrum of human illness. Trans R Soc Trop Med Hyg. 1979; 73(6):630–633. PMID: 575446.

- 5. Meegan JM. Rift Valley fever in Egypt: an overview of the epizootica in 1977 and 1978. Proceedings of workshop on Rift Valley fever; 1981 March 18–21; Herzlia, Israel. p. 100–13.
- Ksiazek TG, Jouan A, Meegan JM, Le Guenno B, Wilson ML, Peters CJ, et al. Rift Valley fever among domestic animals in the recent West African outbreak. Res Virol. 1989 Jan 1; 140:67–77. PMID: 2711047.
- Jupp PG, Kemp A, Grobbelaar A, Leman P, Burt FJ, Alahmedt AM, et al. The 2000 epidemic of Rift Valley fever in Saudi Arabia: mosquito vector studies. Med Vet Entomol. 2002; 16:245–252. PMID: 12243225.
- Shoemaker T, Boulianne C, Vincet MJ, Pezzanite L, Al-Qahtani MM, Al-Mazrou Y, et al. Genetic analysis of viruses associated with emergence of Rift Valley fever in Saudi Arabia and Yemen, 2000–01. Emerg Infect Dis. 2002; 8(12):1415–1420. PMCID: PMC2738516. https://doi.org/10.3201/eid0812. 020195 PMID: 12498657
- Balkhy HH, Memish ZA. Rift Valley fever: an uninvited zoonosis in the Arabian Peninsula. Int J Antimicrob Agents. 2003; 21:153–157. PMID: 12615379.
- Sissoko D, Giry C, Gabrie P, Tarantola A, Pettinelli F, Collet L, et al. Rift Valley fever, Mayotte, 2007–2008. Emerg Infect Dis. 2009; 15(4):568–70. PMCID: PMC2671425. https://doi.org/10.3201/eid1504.081045 PMID: 19331733
- Morvan J, Saluzzo JF, Fontenille D, Rollin PE, Coulanges P. Rift Valley fever on the east coast of Madagascar. Res Virol. 1991; 142(6):475–82. <u>https://doi.org/10.1016/0923-2516(91)90070-J</u>. PMID: 1687082
- 12. Pienaar NJ, Thompson PN. Temporal and spatial history of Rift Valley fever in South Africa: 1950 to 2011. Onderstepoort J Vet Res. 2013; 80(1):1–13. https://doi.org/10.4102/ojvr.v80i1.384.
- Archer BN, Thomas J, Weyer J, Cengimbo A, Landoh DE, Jacobs C, et al. Epidemiologic investigations into outbreaks of Rift Valley fever in humans, South Africa, 2008–2011. Emerg Infect Dis. 2013 Dec; 19 (12):1918. PMCID: PMC3840856.
- Glancey MM, Anyamba A, Linthicum KJ. Epidemiologic and environmental risk factors of Rift Valley fever in southern Africa from 2008 to 2011. Vector Borne Zoonotic Dis. 2015; 15(8):502–11. PMCID: PMC4545538. https://doi.org/10.1089/vbz.2015.1774 PMID: 26273812
- Linthicum KJ, Davies FG, Kairo A, Bailey CL. Rift Valley fever virus (family Bunyaviridae, genus Phlebovirus). Isolations from Diptera collected during an inter-epizootic period in Kenya. Epidemiol Infect. 1985(1):197–209. PMCID: PMC2129511.
- Romoser WS, Oviedo MN, Lerdthusnee K, Patrican LA, Turell MJ, Dohm DJ et al. Rift Valley fever virus-infected mosquito ova and associated pathology: possible implications for endemic maintenance. Res Rep Trop Med. 2011; 2:121–7. https://doi.org/10.2147/RRTM.S13947 PMID: 30881185
- Fafetine J, Neves L, Thompson PN, Paweska JT, Rutten VP, Coetzer JA. Serological evidence of Rift Valley fever virus circulation in sheep and goats in Zambezia Province, Mozambique. PLoS Negl Trop Dis. 2013; 7(2):e2065. PMCID: PMC3585041. <u>https://doi.org/10.1371/journal.pntd.0002065</u> PMID: 23469300
- Niklasson B, Liljestrand J, Bergström S, Peters CJ. Rift Valley fever: a sero-epidemiological survey among pregnant women in Mozambique. Epidemiol Infect. 1987; 99(2):517–22. PMCID: PMC2249273. PMID: 3678406
- Fafetine JM, Coetzee P, Mubemba B, Nhambirre O, Neves L, Coetzer JA, et al. Rift Valley fever outbreak in livestock, Mozambique, 2014. Emerg Infect Dis. 2016; 22(12):2165. PMCID: PMC5189134. https://doi.org/10.3201/eid2212.160310 PMID: 27869589
- Moiane B, Mapaco L, Thompson P, Berg M, Albihn A, Fafetine J. High seroprevalence of Rift Valley fever phlebovirus in domestic ruminants and African Buffaloes in Mozambique shows need for intensified surveillance. Inf Ecol Epidemiol. 2017; 7(1):1416248. PMCID: PMC5757227.
- Kokernot RH, Heymann CS, Muspratt J, Wolstenholme B. Studies on arthropod-borne viruses of Tongaland. V. Isolation of Bunyamwera and Rift Valley fever viruses from mosquitoes. S Afr J Med Sci. 1957 Oct; 22(2–3):71–80. PMID: 13506705.
- 22. Kokernot RH, Smithburn KC, Kluge E. Neutralizing antibodies against arthropod-borne viruses in the sera of domestic quadrupeds ranging in Tongaland, Union of South Africa. Ann Trop Med Parasitol. 1961 Apr 1; 55(1):73–85. PMID: 13757664.
- Smithburn KC, Kokernot RH, Heyman CS, Weinbren MP, Zentkowsky D. Neutralizing antibodies for certain viruses in the sera of human beings residing in Northern Natal. S Afr Med J. 1959 Jul 4; 33 (27):555–61. PMID: 13675877.
- McIntosh BM, Jupp PG, De Sousa J. Further isolations of arboviruses from mosquitoes collected in Tongaland, South Africa, 1960–1968. J Med Entomol. 1972 Apr 28; 9(2):155–9. PMID: 4402531.

- McIntosh BM, Jupp PG, Dos Santos I, Barnard BJ. Vector studies on Rift Valley fever virus in South Africa. S Afr Med J. 1980; 58(3):127–32. PMID: 6105722.
- McIntosh BM, Jupp PG, Dos Santos I, Rowe AC. Field and laboratory evidence implicating *Culex zom-baensis* and *Aedes circumluteolus* as vectors of Rift Valley fever virus in coastal South Africa. S Afr J Sci. 1983 Feb; 79:61–4.
- Rostal MK, Evans AL, Sang R, Gikundi S, Wakhule L, Munyua P, et al. Identification of potential vectors of and detection of antibodies against Rift Valley fever virus in livestock during interepizootic periods. Am J Vet Res. 2010 May; 71(5):522–6. https://doi.org/10.2460/ajvr.71.5.522 PMID: 20433377.
- Sumaye RD, Geubbels E, Mbeyela E, Berkvens D. Inter-epidemic transmission of Rift Valley fever in livestock in the Kilombero River Valley, Tanzania: a cross-sectional survey. PLoS Negl Trop Dis. 2013 Aug; 7(8):e2356. PMCID: PMC3738442. https://doi.org/10.1371/journal.pntd.0002356 PMID: 23951376
- Britch SC, Binepal YS, Ruder MG, Kariithi HM, Linthicum KJ, Anyamba A, et al. Rift Valley fever risk map model and seroprevalence in selected wild ungulates and camels from Kenya. PLoS One. 2013 Jun 28; 8(6):e66626. PMCID: PMC3695998. https://doi.org/10.1371/journal.pone.0066626 PMID: 23840512
- Fagbo S, Coetzer JA, Venter EH. Seroprevalence of Rift Valley fever and lumpy skin disease in African buffalo (Syncerus caffer) in the Kruger National Park and Hluhluwe-iMfolozi Park, South Africa. J S Afr Vet Assoc. 2014 Oct 16; 85(1):e1–7. https://doi.org/10.4102/jsava.v85i1.1075 PMID: 25686252.
- Olaleye OD, Tomori O, Schmitz H. Rift Valley fever in Nigeria: infections in domestic animals. Rev Sci Tech. 1996 Sep 1; 15(3):937–46. PMID: <u>9025143</u>
- LaBeaud AD, Muchiri EM, Ndzovu M, Mwanje MT, Muiruri S, Peters CJ, et al. Interepidemic Rift Valley fever virus seropositivity, northeastern Kenya. Emerg Infect Dis. 2008 Aug; 14(8):1240–46. PMCID: PMC2600406. https://doi.org/10.3201/eid1408.080082 PMID: 18680647
- Lichoti JK, Kihara A, Oriko AA, Okutoyi LA, Wauna JO, Tchouassi DP, et al. Detection of Rift Valley fever virus interepidemic activity in some hotspot areas of Kenya by sentinel animal surveillance, 2009– 2012. Vet Med Int. 2014; 2014:1–9. PMCID: PMC4147350.
- Mbotha D, Bett B, Kairu-Wanyoike S, Grace D, Kihara A, Wainaina M, et al. Inter-epidemic Rift Valley fever virus seroconversions in an irrigation scheme in Bura, south-east Kenya. Transbound Emerg Dis. 2018 Feb; 65(1):e55–e62. https://doi.org/10.1111/tbed.12674 PMID: 28710814.
- Bruton MN. Introduction. In: Bruton MN, Cooper KH, editors. Studies on the ecology of Maputoland. Grahamstown: Rhodes University; 1980. p. xvii–xix.
- Jozini Local Municipality Integrated Development Plan 2017/18–2021/22 [Internet]. Jozini Municipality; c2017 [cited 2018 Sep 16]. Available from: www.jozini.gov.za/phocadownload/idpbudget/June2017/ jozini%20municipality%2017_18%20%20idp.pdf.
- Moll EJ. Terrestrial plant ecology. In: Bruton MN, Cooper KH, editors. Studies on the Ecology of Maputaland. Grahamstown: Rhodes University; 1980. p. 52–68.
- Rogers KH. The vegetation of the Pongolo floodplain: distribution and utilization. In: Bruton MN, Cooper KH, editors. Studies on the Ecology of Maputaland. Grahamstown: Rhodes University; 1980. p. 69–77.
- River Freshwater Ecosystem Priority Areas (FEPAs) (River_FEPAs.shp) [Internet]. South African National Biodiversity Institute; c2016 [cited 2018 May 30]. Available from: <u>www.sasdi.net/metaview.</u> aspx?uuid=0dadbc5046456801420ebea919939f99.
- 40. Wetland Freshwater Ecosystem Priority Areas (NFEPA_Wetlands.shp) [Internet]. South African National Biodiversity Institute; c2016 [cited 2018 May 30]. Available from: www.sasdi.net/metaview.aspx?uuid=330feba934dbd53fc832cee55768e188.
- **41.** Ezemvelo Protected Area boundary (ekznw_pabnd_2015_wdd.zip) [Internet]. Biodiversity Conservation Planning Division, Ezemvelo KZN Wildlife; c2015 [cited 2018 Oct 29]. Available from: http://www.sasdi.net/metaview.aspx?uuid=78e859b1c3161984b3e02739c58b3241.
- 42. Thrusfield MV. Veterinary Epidemiology. 3rd ed. Oxford: Blackwell Science; 2007.
- **43.** Bennett S, Woods T, Liyanage WM, Smith DL. A simplified general method for cluster sample surveys of health in developing countries. World Health Stat Q. 1991; 44(3):98–106. PMID: 1949887.
- Paweska JT, Burt FJ, Anthony F, Smith SJ, Grobbelaar AA, Croft JE, et al. IgG-sandwich and IgM-capture enzyme-linked immunosorbent assay for the detection of antibody to Rift Valley fever virus in domestic ruminants. J Virol Methods. 2003 Nov; 113(2):103–12. PMID: 14553896.
- World Organisation for Animal Health (OIE). Rift Valley fever. In: Manual of Diagnostic Tests and Vaccines for Terrestrial Animals. 7th ed. Paris: OIE; 2012. ch. 2.1.14.
- 46. Karber G. 50% end point calculation. Arch Exp Pathol Pharmakol. 1931; 162:480-3.

- Vandormael A, Dobra A, Bärnighausen T, de Oliveira T, Tanse F. Incidence rate estimation, periodic testing and the limitations of the mid-point imputation approach. Int J Epidemiol. 2018 Feb; 47(1):236– 45. PMCID: PMC5837439. https://doi.org/10.1093/ije/dyx134 PMID: 29024978
- Swanepoel R. Observations on Rift Valley fever in Zimbabwe. Contrib Epidemiol Biostat. 1981; 3:83– 91.
- Nanyingi MO, Muchemi GM, Thumbi SM, Ade F, Onyango CO, Kiama SG, et al. Seroepidemiological survey of Rift Valley fever virus in ruminants in Garissa, Kenya. Vector Borne Zoonotic Dis. 2017 Feb; 17(2):141–6. https://doi.org/10.1089/vbz.2016.1988 PMID: 27929928.
- 50. Eisa M. Preliminary survey of domestic animals of the Sudan for precipitating antibodies to Rift Valley fever virus. Epidemiol Inf. 1984 Dec; 93(3):629–37. PMCID: PMC2129467.
- Rissmann M, Eiden M, El Mamy BO, Isselmou K, Doumbia B, Ziegler U, et al. Serological and genomic evidence of Rift Valley fever virus during inter-epidemic periods in Mauritania. Epidemiol Inf. 2017 Apr; 145(5):1058–68. PMID: 28029091.
- Cêtre-Sossah C, Pédarrieu A, Guis H, Defernez C, Bouloy M, Favre J, et al. Prevalence of Rift Valley fever among ruminants, Mayotte. Emerg Inf Dis. 2012 June; 18(6):972–5. PMCID: PMC3358149.
- Umuhoza T, Berkvens D, Gafarasi I, Rukelibuga J, Mushonga B, Biryomumaisho S. Seroprevalence of Rift Valley fever in cattle along the Akagera-Nyabarongo rivers, Rwanda. J S Afr Vet Assoc. 2017 Jan 20; 88(1):e1–5. https://doi.org/10.4102/jsava.v88i0.1379 PMID: 28240033.
- 54. Lagerqvist N, Moiane B, Mapaco L, Fafetine J, Vene S, Falk KI. Antibodies against Rift Valley fever virus in cattle, Mozambique. Emerg Inf Dis. 2013 Jul; 19(7):1177–9. PMCID: PMC3713994.
- 55. Blomström AL, Scharin I, Stenberg H, Figueiredo J, Nhambirre O, Abilio A, et al. Seroprevalence of Rift Valley fever virus in sheep and goats in Zambézia, Mozambique. Infect Ecol Epidemiol. 2016 Jul 4; 6 (1):31343. PMCID: PMC4933789.
- Boussini H, Lamien CE, Nacoulma OG, Kaboré A, Poda G, Viljoen G. Prevalence of Rift Valley fever in domestic ruminants in the central and northern regions of Burkina Faso. Rev Sci Tech. 2014 Dec; 33 (3):893–901. PMID: 25812213.
- 57. Soumare B, Tempia S, Cagnolati V, Mohamoud A, Van Huylenbroeck G, Berkvens D. Screening for Rift Valley fever infection in northern Somalia: a GIS based survey method to overcome the lack of sampling frame. Vet Microbiol. 2007 Apr 15; 121(3–4):249–56. https://doi.org/10.1016/j.vetmic.2006.12.017 PMID: 17257783.
- Vogel C, Van Zyl K. Drought: In search of sustainable solutions to a persistent, "wicked" problem in South Africa. In: Salzmann N, Huggel C, Nussbaumer SU, Ziervogel G, editors. Climate Change Adaptation Strategies–An Upstream-downstream Perspective. Heidelberg: Springer; 2016. p. 195–211.
- 59. Porta M, editor. A Dictionary of Epidemiology. 6th ed. Oxford: Oxford University Press; 2014.
- Beechler BR, Bengis R, Swanepoel R, Paweska JT, Kemp A, Van Vuren PJ, et al. Rift Valley fever in Kruger National Park: Do buffalo play a role in the inter-epidemic circulation of virus? Transbound Emerg Dis. 2015 Feb; 62(1):24–32. https://doi.org/10.1111/tbed.12197 PMID: 24330522.
- Gudo ES, Pinto G, Weyer J, Le Roux C, Mandlaze A, José AF, et al. Serological evidence of Rift Valley fever virus among acute febrile patients in Southern Mozambique during and after the 2013 heavy rainfall and flooding: implication for the management of febrile illness. Virol J. 2016 Jun; 13:96–102. https://doi.org/10.1186/s12985-016-0542-2 PMID: 27278404.